\documentstyle[12pt]{article}
\def\baselinestretch{1.30}
\topmargin = -0.9cm
\oddsidemargin=-0.5cm
\textheight=21cm
\textwidth=16cm

\begin{document}

\pagenumbering{roman}

       \centerline{\bf  Information Sheet}
     \medskip

     \centerline{\it Math 104, Section 2,   Spring 1997 }

\bigskip




\bigskip

\noindent
{\bf Instructor:}  

Prof. John Cao,  Room 221 CCMB, 631-8847 (office), 
 E-mail address: jcao@jcao.math.nd.edu

\noindent
{\bf Textbook:} 

 L. J. Goldstein, D. I. Schneider and M. J. Siegel:  {\it ``Finite 
Mathematics and Its Applications"}, Prentice Hall,  5th Edition, 1995.

\noindent
   The course will cover nearly all of chapters 1, 2, 5-8 and 10.

              


\noindent
{\bf Time to meet:}   MWF 11:45-12:35,  Pasq Center, Room 108.
                      

\noindent
{\bf Office Hours:} 

Prof. Cao: Wed 3:00-4:00 pm, Room 221 CCMB, or by appointment. 

       
\noindent
{\bf Exams:}

  
\noindent
Exam I,  Thursday, Feb. 13, 8:00 am - 9:15 am;  Room 126 DBRT \newline 
Exam II, Thursday, March 20,  8:00 am - 9:15 am;  Room 126 DBRT  \newline 
 Exam III, Thursday, April 24, 8:00 am - 9:15 am;  Room 126 DBRT  \newline 
 Final Exam, Thursday, May 8,  1:45pm - 3:45 pm.  


  The best way
to prepare exams is to work problems and work problems and work more 
problems. It is important to do homework as the course goes along.
If you start working and studying just a few days before an exam, there 
is no way you can reach the level you would be at if you had done some each week and then reviewed for the exam. 

 



        If there 
are conflicts with exams, please let Professor Cao know during
the first week of classes so that we can try to resolve them.
 

\noindent
{\bf How things count:} 

Homework and quizzes $ 25 \%$ (150/600); Three midterm exams: $ 3 \times 16.66 \% 
= 50 \% $ (300/600); Final Exam $ 25 \% $ (150/600). The total grades will
be 600.


\noindent
{\bf Homework  Assignment Policy:} 

 One third credits will be given each time if you hand in 
your homework  on time.  No later homework 
 will be accepted. Homework 
sets are due every Monday and Friday's lectures. 



\medskip














              


  

  











\end{document}
